Author’s response to reviews

Title: Medical Students' Perception on Fecal Microbiota Transplantation

Authors:

Petru Madar (petrumadar@gmail.com)
Oana Petre (oanapetre@psychology.ro)
Adriana Baban (adrianababan@psychology.ro)
Dan Dumitrascu (dumitrascu@umfcluj.ro)

Version: 2 Date: 23 Aug 2019

Author’s response to reviews:

Dear Editor,

Thank you for the suggestions on our revised manuscript MEED-D-19-00085R1.

We introduced a few sentences to reply to the reviewer's comment:

"The only remaining point is to consider some commentary on how the results of this survey are generalisable to medical educators, i.e. inform those in charge of the medical curriculum as to what new technologies to include, when (i.e how soon after they are identified), and how best to do that?"

We hope that this addition is what the reviewer expected and that our manuscript is improved with this comment.

Thank you for processing our manuscript.

Best regards,

Prof. D. L. Dumitrascu on behalf of the authors